Author's response to reviews

Title: Age of donor alters the effect of cyclic hydrostatic pressure on production by human macrophages and osteoblasts of sRANKL, OPG and RANK

Authors:

Carol CE Evans (carol.evans@man.ac.uk)
Sarah S Mylchreest (sarah.lewis@man.ac.uk)
Glynne JG Andrew (Glynne.Andrew@nww-tr.wales.nhs.uk)

Version: 8 Date: 22 February 2006

Author's response to reviews: see over
The full stop has been removed from the end of the title in the MS and the highlighting/tracking removed. Tables 2a & 2b have been altered to table 2 & 3 and the text altered to reflect this. The titles have been removed from all the Figures and as much white space as possible removed from around the Figures.